\documentclass{article}
\usepackage{fullpage,amsmath}
\usepackage{hyperref}
\usepackage{amssymb}
\usepackage{listings}
\usepackage{amsthm}



\title{CSC165, Summer 2014\\
Lecture notes: Week 1}
\renewcommand{\today}{~}
\hypersetup{pdfpagemode=Fullscreen,
  colorlinks=true,
  linkfileprefix={}}
\begin{document}



\maketitle



\section{A little bit of \LaTeX}
Look inside the \textbf{tex} file to see how to do these things!

Here's how you make a table:

\begin{tabular}{|c|c|} 							\hline
  	CSC108 			& 	CSC165		\\ 	\hline
  	Programming 		& 	Logic 	 		\\ 	\hline
\end{tabular}




Here's how you make a list:
\begin{enumerate}
	\item May
	\item June
	\item July
	\item August
\end{enumerate}



Here's how you make an unordered list:
\begin{itemize}
	\item BA1170
	\item BA1190
\end{itemize}

Here's how you can you can say that there exists a rational number greater than $10$ (note that math mode dollar signs):

$\exists x, x > 10, x \in \mathbb{R}$

Here's how you can say that there is no natural number smaller than $-100$:

$\not \exists n, n < -100, n \in \mathbb{N}$

Here's how you can say that there is no whole number that's not 0 that equals to $42$ (this happens to be false, but you can still \textit{say} it!

$\not \exists n, n = 42, n \in \mathbb{Z^*}$

\end{document}